# Matrix Metalloproteinase 9 Production by Monocytes is Enhanced by TNF and Participates in the Pathology of Human Cutaneous Leishmaniasis



Taís M. Campos<sup>1</sup>, Sara T. Passos<sup>1,2</sup>, Fernanda O. Novais<sup>3</sup>, Daniel P. Beiting<sup>3</sup>, Rúbia S. Costa<sup>1</sup>, Adriano Queiroz<sup>1</sup>, David Mosser<sup>4</sup>, Phillip Scott<sup>3</sup>, Edgar M. Carvalho<sup>1,2</sup>, Lucas P. Carvalho<sup>1,2,5</sup>\*

Serviço de Imunologia, Universidade Federal da Bahia, Salvador, Bahia, Brazil, 2 Instituto Nacional de Ciências e Tecnologia – Doenças Tropicais, Salvador, Bahia, Brazil,
Department of Pathobiology, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America, 4 Department of Cell Biology and Molecular Genetics,
University of Maryland, College Park, Maryland, United States of America, 5 Instituto de Ciências da Saúde, Universidade Federal da Bahia, Salvador, Bahia, Brazil

# Abstract

*Introduction:* Cutaneous leishmaniasis (CL) due to *L.braziliensis* infection is characterized by a strong inflammatory response with high levels of TNF and ulcer development. Less attention has been given to the role of mononuclear phagocytes to this process. Monocytes constitute a heterogeneous population subdivided into classical, intermediate and non-classical, and are known to migrate to inflammatory sites and secrete inflammatory mediators. TNF participates in the induction of matrix metalloproteinases (MMPs). MMP-9 is an enzyme that degrades basal membrane and its activity is controlled by the tissue inhibitor of metalloproteinase.

*Methods:* Mononuclear cells were obtained from *ex-vivo* labeling sub-populations of monocytes and MMP-9, and the frequency was determined by flow cytometry. Culture was performed during 72 hours, stimulating the cells with SLA, levels of MMP-9 and TIMP-1 in the supernatants were determined by ELISA.

*Results:* We observed that cells from CL lesions secrete high amounts of MMP-9 when compared to healthy subjects. Although MMP-9 was produced by monocytes, non-classical ones were the main source of this enzyme. We also observed that TNF produced in high level during CL contributes to MMP-9 production.

*Conclusions:* These observations emphasize the role of monocytes, TNF and MMP-9 in the pathogenesis of *L. braziliensis* infection.

Citation: Campos TM, Passos ST, Novais FO, Beiting DP, Costa RS, et al. (2014) Matrix Metalloproteinase 9 Production by Monocytes is Enhanced by TNF and Participates in the Pathology of Human Cutaneous Leishmaniasis. PLoS Negl Trop Dis 8(11): e3282. doi:10.1371/journal.pntd.0003282

Editor: Mary Ann McDowell, University of Notre Dame, United States of America

Received January 30, 2014; Accepted September 19, 2014; Published November 13, 2014

**Copyright:** © 2014 Campos et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** This work was supported by National Institutes of Health [grant Al088650] and Conselho Nacional de Pesquisa Instituto Nacional de Ciência e Tecnologia – Doenças Tropicais [grant 573839/2008-5]. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* Email: carvalholp@ig.com.br

# Introduction

Human cutaneous leishmaniasis (CL) caused by *Leishmania* braziliensis infection is characterized by the presence of one or more ulcerated lesions with raised borders and few parasites [1]. Early after infection, most patients develop lymphadenopathy, followed by the appearance of a papule at the bite site, which subsequently becomes an ulcerated lesion. These lesions are composed of a robust inflammatory infiltrate including the presence of T and B lymphocytes, mononuclear phagocytes and plasma cells [2]. It is well known that both CD4+ and CD8+ T cells have important roles in the control of *Leishmania* parasites replication [3]. During *L. braziliensis* infection these cells are activated and TNF and IFN- $\gamma$  are produced in high levels both by peripheral blood mononuclear cells (PBMC) and at lesion site of CL patients. However, this response can also lead to tissue damage and development of the ulcer [4].

In contrast to the role of T cells in the pathogenesis of L. braziliensis infection less attention has been given to the contribution of mononuclear phagocytes to the inflammatory process and tissue damage observed in CL. In spite of the presence of tissue resident mononuclear phagocytes, circulating monocytes migrate to the infection site. In the tissue, they can differentiate into macrophages and dendritic cells (DCs), which are the main cell types parasitized, by *Leishmania*. It was recently proposed that in humans circulating monocytes are a heterogeneous population based on the surface expression of CD14 and CD16 [5]. The monocyte subsets are subdivided into classical (CD14++CD16-), intermediate (CD14++CD16+) and nonclassical (CD14+CD16++) [5,6]. It has been shown that CD16+ monocytes are able to produce high levels of TNF and increased frequency of these monocytes is associated with the immunopathogenesis of inflammatory diseases, such as arthritis and sepsis [7,8,9].

# **Author Summary**

To examine the participation of MMP-9 in the pathogenesis of L. braziliensis infection, we realized a cross-sectional study with CL patients in an early phase of the disease or with a classical ulcer, and healthy controls. We evaluated the frequency of MMP-9 in monocyte subsets and its mechanism of production. Our results showed that monocytes were the major cells producing MMP-9. The MMP-9 production by CL patients was presented in higher levels when compared with healthy subjects and early cutaneous leishmaniasis (ECL) patients, and the levels of MMP-9 inhibitor, TIMP-1, were lower in CL patients when compared to healthy subjects. The production of MMP-9 was enhanced by TNF, a cytokine associated with tissue damage in CL patients. Thus, therapeutic modulation of MMP-9 may be a useful approach for improving disease outcome in L. braziliensis patients.

TNF is an inflammatory cytokine produced in high levels in CL patients [4]. TNF participates in the inflammatory process through the induction of nitric oxide, necrosis, citotoxicity and expression of matrix metalloproteinases (MMPs) [10,11,12]. MMPs are zincdependent enzymes that degrade extracellular matrix proteins and are functionally classified according to the specificities of the substrates they degrade [13]. MMP-9 is involved in degradation of collagen type IV, the major component of basal membrane present at skin [14]. While MMPs are necessary for successful eradication of infection by stimulating the migration of effector cells to the inflammatory site, high production of these molecules may induce pathology [15]. An imbalanced production of MMPs and its natural regulator, tissue inhibitor of metalloproteinases (TIMPs) occurs in a variety of diseases where tissue damage occurs [16,17,18]. L. braziliensis-infected macrophages secrete MMP-9, and its production is increased in patients with mucosal leishmaniasis, the more severe form of L. braziliensis infection [19]. Thus, while the factors that induce the breakdown of the basal membrane leading to the development of skin ulcer in leishmaniasis are not yet elucidated, we hypothesize that the lesion formation is due to breakdown dysregulation of the basal membrane caused by imbalance in levels of MMP-9 and TIMP-1, together with other factors, such as cell recruitment and edema, [20].

In the present study we examined the participation of MMP-9 in the pathogenesis of *L. braziliensis* infection. CL patients in an early phase of the disease or with a classical ulcer were evaluated. Initially, we showed that MMP-9 genes were highly expressed in the CL skin lesions. We then evaluated the production of MMP-9 and its inhibitor, TIMP-1, and identified an imbalance between MMP-9 and TIMP-1 ratio in CL patients. Monocytes were the main source of the enzyme. Finally, we found that high levels of TNF produced during the disease contribute to the up-regulation of MMP-9 synthesis in CL.

### Materials and Methods

## Patients

Participants of the present study includes 19 early cutaneous leishmaniasis (ECL) and 85 CL patients from the *L. braziliensis* transmission area of Corte de Pedra, Bahia state, Brazil, and 29 healthy subjects (HS) living in areas not exposure to *Leishmania*. ECL patients were characterized by the presence of a lymphadenopathy or lymphadenopathy accompanied by a papule or an exoulcerative lesion. Diagnosis for CL was performed by a positive

parasite culture or PCR as previously described [21]. All patients were evaluated before treatment.

# Microarray-based expression profiling of human lesions

For whole genome expression microarray, lesion biopsies preserved in RNAlater (Qiagen) were homogenized using a rotor-stator and RNA was isolated using the RNeasy Plus kit (Qiagen). Biotin-labeled complementary RNA (cRNA) was generated using the Illumina TotalPrep RNA amplification kit (Ambion). RNA and cRNA quality were assessed on a BioAnalyzer (Agilent). Illumina HumanHT-12 version 4 expression beadchips were hybridized with cRNA from 26 L. braziliensis lesion biopsies and 10 biopses collected from uninfected donors. Arrays were scanned using a BeadStation 500GX and raw image files were processed using GenomeStudio v1.8 software (Illumina). Data was variance stabilized, robust-spline normalized, and quality control analysis carried out using the Lumi package [22] in Bioconductor/R. Differential expression analysis of the data using linear models and empirical Bayes methods [23] was carried out using the Limma package [24]. Data was deposited on the Gene Expression Omnibus (GEO) database for public access (GSE#GSE43880). Heat map tools available on GenePattern [25] were used to graphically display differentially regulated genes in Figure 1A.

## Soluble Leishmania Antigen (SLA) preparation

SLA was prepared with an isolate of *L. braziliensis* as previously described [26]. Briefly, promastigotes ressuspended in lysis solution (Tris, HCL, EDTA and leupeptin) were immersed in liquid nitrogen, and thawed at 37°C. After freezer-thaw procedure, they were sonicated and the disrupted parasites were centrifuged at 14,000G. The supernatant was filtered and assayed for protein concentration. SLA was used at a concentration of (5  $\mu$ g/ml).

#### Flow cytometry

Flow cytometry was performed as previously described [27]. Briefly, PBMC were ressuspended in saline and adjusted to a concentration of  $0.5 \times 10^6$  cells/ml. For *ex-vivo*, cell surface staining, we incubated cells with monoclonal antibodies anti-CD4 (APC), anti-CD8 (PE), anti-CD14 (APC) and anti-CD16 (PE) (BD bioscience), for 15 minutes, washed by centrifugation twice and fixed with 2% paraformaldehyde. For intracellular staining, cells were ressuspended in Perm/Wash (BD Cytofix/Cytoperm Plus – BD bioscience) for 15 minutes and intracellular labeling was performed using monoclonal antibody anti-MMP-9 (FITC) (BD bioscience) for 30 minutes. Cells were acquired on a FACS Canto II and analysis was done using Flowjo software (Tristar).

# Cultures of peripheral blood mononuclear cells and biopsies

Peripheral blood mononuclear cells (PBMC) were isolated from heparinized venous blood by Ficoll-Hypaque (GE Healthcare Bio-Sciences AB, Sweden) gradient centrifugation. After washing three times in 0.9% NaCl, PBMC were adjusted to  $3 \times 10^6$  cells/ml in 1 ml of RPMI-1640 (Gibco Laboratories, Grand Island, NY, USA) supplemented with 10% fetal bovine serum (FBS) (Gibco Laboratories, South America Invitrogen), 10 IU/ml penicillin and 100 µg/ml streptomycin. Cells were placed on 24 wells plates and incubated for 24 or 72 hours in the presence or absence of SLA (5 µg/ml) or recombinant TNF (5 ng/ml), or anti-TNF antibody (10 µg/ml) (R&D systems, Minneapolis, MN), as indicated in figures.



**Figure 1. Lesions of CL patients produce MMP-9.** *A*, Heatmap showing expression of MMPs and TIMP-1 from microarray profile. Biopsies from CL patients (n = 26) and normal skin (n = 10) were obtained and unbiased microarray was performed on biopsies mRNA. Average fold change (FC) for each gene in lesion samples relative to normal skin controls is shown. *B*, Ratio between MMP-9 and TIMP-1 genes expression. Biopsies from CL patients (n = 26) and normal skin (n = 10) were obtained and unbiased microarray was performed on biopsies mRNA. Average fold change (FC) for each gene in lesion samples relative to normal skin controls is shown. *B*, Ratio between MMP-9 and TIMP-1 genes expression. Biopsies from CL patients (n = 26) and normal skin (n = 10) were obtained and unbiased microarray was performed on biopsies mRNA. *C*, MMP-9 levels in biopsies culture supernatants from CL patients (n = 6) and healthy subjects (HS) (n = 5), determined by ELISA after the biopsies been cultured for 12 h in absence of stimuli. \*\*p<0.005. doi:10.1371/journal.pntd.0003282.g001

Biopsies from *L. braziliensis* patients and HS were cultured in complete RPMI media without stimuli. Tissue from CL patients and HS were cultured in RPMI for 12 hours. Supernatants from PBMC and biopsies were collected and stored at  $-70^{\circ}$ C. The Levels of MMP-9, TIMP-1 (BD Biosciences, San Diego, CA, USA) and TNF (R&D Systems, Minneapolis, MN) were measured by ELISA according to the manufactures instructions. The results are expressed in pg/ml.

## Statistical analysis

Mann-Whitney was used to compare HS and CL groups; Kruskal-Wallis test (nonparametric test) was used to compare the ECL, CL and HS groups; Wilcoxon matched pair test (paired and nonparametric t test) was used to analyze PBMC cultures in different conditions within the same group of individuals. Comparisons were considered statistically significant when p< 0.05. All p values represented are two-sided.

#### Ethical considerations

This work was approved by the Ethics and Research Committee from Federal University of Bahia. All subjects provided witted informed consent; in case of illiterate subjects, a thumb print plus signature of an independent witness were used.

# Results

## MMP-9 is produced at lesion site from CL patients

MMPs are enzymes that degrade extracellular matrix and in high levels MMPs may cause tissue damage [15]. Exaggerated inflammatory responses leads to tissue damage and ulcer development in CL [4]. To determine whether MMPs were expressed in CL lesions, we performed a whole genome expression profile from lesions of CL patients (n = 26) and compared that to normal skin (n = 10). The results showed that several MMPs, including MMP-9, had increased expression over normal skin (Figure 1A). We also assessed the expression of TIMP-1 in the skin and although TIMP-1 expression was increased in *L. braziliensis* infected skin over healthy skin, the ratio MMP-9/TIMP-1 expression was higher in CL skin when compared to normal tissue (Figure 1A and B). Because MMP-9 is known to be involved in basal membrane disruption, a process that precedes ulcer development in CL, we tested whether MMP-9 protein was produced in the lesion of CL patients and healthy subjects. To address that we cultured whole biopsies in the absence of stimulus for 12 hours and determined MMP-9 levels on culture supernatants by ELISA. Cells from biopsies from CL patients produced significantly more MMP-9 than those from healthy subjects (HS) (Figure 1C). These results show an imbalance between MMP-9 and TIMP-1 expression, suggesting that MMP-9 participate in lesion development in CL.

# Imbalanced production of MMP-9 and TIMP-1 in CL patients

TIMPs are glycoproteins that inhibit MMPs and TIMP-1 specifically regulates the activity of MMP-9. Therefore, our next step was to determine the production MMP-9 and TIMP-1 by PBMC from HS and CL patients in the early and late phase of the disease. To do so, PBMC were stimulated in vitro with SLA or left untreated (media) for 72 hours and the production of MMP-9 and TIMP-1 were determined in the supernatants by ELISA. Our results show that CL patients produced high levels of MMP-9 and low levels of TIMP-1 when compared to HS (Figure 2A and 2B). Interestingly, PBMCs from CL patients produced high levels of MMP-9 even in the absence of stimulus, indicating that those cells are prone to produced MMP-9 when still in CL patient. A change in the ratio of MMPs and TIMPs may cause excessive degradation of the extracellular matrix and consequently tissue damage [28]. Therefore, we next sought to determine the ratio between MMP-9 production and TIMP-1. The high ratio MMP-9/TIMP-1 in cultures from CL individuals indicated an imbalance in the production of these enzymes in L. braziliensis infected individuals when compared to HS (Figure 2C).

# MMP-9 is mainly produced by monocytes from CL patients

It has been documented that leucocytes are the main source of MMP-9 [29,30] and our results show that PBMCs from CL patients secrete high levels of MMP-9 in response to SLA. To further characterize the source of MMP-9 in CL patients, using flow cytometry we determined the *ex-vivo* production of MMP-9



**Figure 2. CL patients have high levels of MMP-9 and low levels of TIMP-1 when compared with healthy subjects.** PBMC were obtained from healthy subjects (HS) (n = 12), early CL (ECL) (n = 19) and CL patients (n = 18), and levels of MMP-9 (A) and TIMP-1 (B), determined by ELISA in culture supernatants of PBMCs upon stimulation with soluble *Leishmania* antigen (SLA) (5 µg/ml) for 72 hours. C, Ratio between MMP-9 and TIMP-1 in culture supernatants of PBMCs upon stimulation with SLA. The box-and-wickers plot shows the minimum, first quartile, median, third quartile, and maximum values of a set of data. The wickers represent error bars and those represent minimum and maximum values, indicating variability outside the upper and lower quartiles. Data shown from a representative experiment of five performances. \*p<0.05; \*\*p<0.005; \*\*\*p<0.005. doi:10.1371/journal.pntd.0003282.g002

by CD4+ and CD8+ T cells, and monocytes based on CD14 expression. Our results show that monocytes are the main source of this enzyme in HS and patients with CL (Figure 3A and B).

Recently, three monocyte subsets have been described based on the expression of CD14 and CD16, subdividing them into classical, intermediate and non-classical monocytes [5]. In our study the gate strategy to access the monocyte subsets was defined based on CD14 and CD16 expression. To evaluate the contribution of monocyte subsets to MMP-9 production we performed intracellular *ex-vivo* staining of PBMC from patients with ECL, CL and HS which were then analyzed by flow cytometry. The frequency of MMP-9 in each subsets of monocyte



**Figure 3. Monocytes are important source of MMP-9 in CL.** PBMC were obtained from healthy subjects (HS) and CL patients and intracellular labeling to MMP-9 was performed *ex-vivo*. The frequency of CD4+, CD8+ and CD14+ cells positive to MMP-9 was determined by flow cytometry. The gate strategy was done using all minus one staining. *A*, representative plots from HS (n = 5). *B*, representative plots from CL patients (n = 5). doi:10.1371/journal.pntd.0003282.g003



**Figure 4. Monocyte subsets from CL patients produce MMP-9.** PBMC were obtained from healthy subjects (HS) (n = 17), early CL (ECL) (n = 11) and CL patients (n = 28), and staining to CD14, CD16 and MMP-9 was performed. Frequency of monocyte subsets producing MMP-9 was determined *ex-vivo* by intracellular staining. Gate strategy to assess monocyte subsets based on CD14 and CD16 expression and frequency of MMP-9 in each subset of monocytes in HS (A) and CL patients (B). The gate strategy was done based on all minus one staining. *C*, Frequency of MMP-9-producing monocyte subsets from HS, ECL and CL patients. *D*, Mean fluorescence intensity (MFI) of MMP-9 positive cells in monocyte subsets from HS, ECL and CL patients. Data shown from a representative experiment of four performances. \*p<0.05; \*\*p<0.005; \*\*\*p<0.005. doi:10.1371/journal.pntd.0003282.g004

was performed using an isotype control (Figure 4A and B). Our data shows that monocytes from patients with ECL and CL express more MMP-9 than monocytes from HS and that non-classical monocytes (CD14+CD16++) are the major source of MMP-9 in patients with CL (Figure 4C and 4D).

# TNF enhances MMP-9 production in CL patients

TNF is an inflammatory cytokine that contributes to tissue damage by different mechanisms, including by induction of MMP-9 expression [31]. To determine if TNF production contributes to MMP-9 production in CL patients, we first measured TNF levels in supernatants of SLA-stimulated PBMC from HS and CL patients. As previously demonstrated [32], CL patients produced high levels of TNF in response to SLA (Figure 5A). In order to determine if the TNF produced had the ability to increase MMP-9 expression, we added recombinant TNF to PBMC cultures from HS and determined MMP-9 expression in the supernatants by ELISA. Our results show that there was a 5-fold increase in MMP-9 production in unstimulated PBMC cultures from healthy individuals upon exogenous addition of recombinant TNF (Figure 5B). Finally, we asked if TNF was playing a role in MMP-9 production in CL. In order to address this question, we added anti-TNF antibodies to PBMC cultures from HS and CL patients (Figure 5C) and we found that the blockage of TNF inhibits the production of MMP-9 in CL patients. Altogether these data show that TNF, in part, regulates MMP-9 production in CL patients.

# Discussion

CL is characterized by a well-defined ulcer with raised borders that appears a few weeks after the transmission of the parasite by sandflies [33]. One of the first signs of the disease is lymphadenopathy followed by the development of a papular or an exoulcerative lesion at the bite site, 2-3 weeks later. Finally, 1-2 weeks later the classical ulcer is observed [33,34]. The evaluation of the immune response during the phase in which patients has not yet developed the cutaneous ulcer is important to determine factors contributing to disease development. Our group has described the immune response in CL patients and discovered the presence of many cytokines and chemokines that contributes to the maintenance of inflammatory response in these individuals [35]. However, much less has been done to understand the immune response at the early stages of the disease. This knowledge is particularly important, as it may unravel the factors that contribute to disease severity and/or ulcer development. Here, we investigated the contribution of leucocytes to MMP-9 production in CL patients and for the first time, we show how monocytes contribute to the production of MMP-9 in CL.

MMPs mediate several physiological processes, such as cell migration, extracellular matrix degradation and tissue remodeling. Our *in situ* gene expression data shows that MMP-1, MMP-3 and MMP-9 expression were particularly increased in CL over healthy skin. While MMP-1 and MMP-3 are more associated with matrix remodeling, uncontrolled secretion of MMP-9 has been associated



**Figure 5. TNF enhances MMP-9 production in CL patients.** *A*, PBMC from healthy subjects (HS) (n = 12) and CL patients (n = 19) were cultured in the presence or absence of SLA (5 µg/ml) for 72 hours, and TNF concentrations were determined by ELISA on supernatants. *B*, PBMC from HS (n = 5) were cultured in the presence or absence of recombinant TNF (5 ng/ml) for 24 hours and the levels of MMP-9 were determined by ELISA. *C*, PBMC from HS (n = 6) and CL patients (n = 6) was cultured in presence or absence of anti-TNF antibodies (10 µg/ml) for 24 hours. The levels of MMP-9 were determined by ELISA on supernatants. \*p<0.05; \*\*\*p<0.0005. doi:10.1371/journal.pntd.0003282.g005

with pathological processes [28,36,37]. There are a few examples of a role for MMPs in the pathogenesis of Leishmania infection: in a murine model of *L. chagasi* infection, production of MMP-9 by macrophages was associated with tissue damage [38]; high levels of mRNA to MMP-2 were documented in ulcers of CL patients and in macrophages from mucosal leishmaniasis patients; and upon infection with L. braziliensis human macrophages increased secretion of MMP-9 [19]. As MMP-9 degrades type IV collagen, the main component of the basement membrane in the skin, and their activation causes excessive tissue damage by facilitating the migration of inflammatory cells to the infection site [28]. In the present work we found that the MMP-9 gene was expressed during disease when compared to normal skin and we were also able to detect high levels of this enzyme being secreted by cells obtained from L. braziliensis lesions. With these findings we hypothesize that MMP-9 plays a role in ulcer development during CL. Our results led us to determine the levels of MMP-9 produced by PBMC. High levels of MMP-9 were observed in supernatants of PBMC from patients when compared with those from healthy subjects. Because exaggerated production of MMPs are associated with immunopathology [15], TIMPs, the natural inhibitors of these enzymes, play an important role in controlling MMPs production and activity. As expected, the levels of TIMP-1 were increased in healthy subjects when compared to CL patients, revealing an imbalance between MMP-9/TIMP-1 in CL individuals. MMP-9 activity was previously associated with development of mucosal leishmaniasis, the more inflammatory form of the disease with extensive tissue damage, and expression of MMPs were also correlated with therapeutic failure in CL [19,39]. These data confirm the potential role for these enzymes in mediating immunopathology during leishmaniasis. Our documentation that patients in the pre-ulcerative phase of the disease produced lower levels of MMP-9 than patients with classical ulcer, raise the possibility that progression of the disease to the ulcerative phase may be associated with increasing in MMP-9 levels and, consequently, imbalance between MMP-9 and TIMP-1.

Evidence has been accumulated that activation of both CD4+ and CD8+ T cells are associated with tissue damage in CL [40,41,42]. On the other hand, less attention has been given to the role of mononuclear phagocytes in the pathogenesis of CL. Macrophages are the main reservoirs of the Leishmania, responsible for *Leishmania* parasite killing [43] but macrophages also contribute to the production of TNF- $\alpha$  and pro-inflammatory chemokines, which are associated with severe forms of the disease [44]. It was previously reported that different leukocytes contribute to deleterious MMPs secretion in infectious diseases [29,30,45,46]. In a mouse model of toxoplasmosis, in addition to inflammatory monocytes and neutrophils, CD4+ and CD8+ T lymphocytes where also important source of MMP-8 and MMP-10 [47]. In malaria and tuberculosis MMP-9 was predominantly produced by activated monocytes [45,46]. Here we show that in CL, although CD4+ and CD8+ T lymphocytes participate in the pathological immune response through the production of inflammatory mediators or cytolytic activity, monocytes were the main cells producing MMP-9. Neutrophils are another cell type known to produce high levels of MMP-9 in other inflammatory conditions [48], but since in CL the cellular infiltrate is predominantly mononuclear, neutrophils may not be a primary source of MMP9 in these patients [49].

Monocytes are subdivided in three subsets according to the expression of CD14 and CD16, and each subset has distinct roles during the infectious process. Classical monocytes (CD14++ CD16-) have high phagocytic capacity [5,6] and express more anti-microbicidal molecules, like reactive oxygen species, acting as an important line of defense against *L. braziliensis* (Novais and Nguyen et al., unpublished data). In contrast, intermediate monocytes (CD14++CD16+) express more MHC class II and act more effectively as antigen presenting cells than the other subsets of monocytes [5,6]. Controversial studies have been published regarding the function of non-classical monocytes (CD14++CD16++) as some have reported that these cells produce more pro-inflammatory cytokines and are responsible for the

recruitment of cells to the inflammatory sites, and others have documented a regulatory role for them [5,6]. Although in this study all monocyte subsets produced MMP-9, the intensity of expression of this molecule was higher in non-classical monocytes from some individuals. It was previously shown that the expression of CD16 in monocytes is associated with high production of MMP-9 and increased frequency of CD16+ monocytes has been documented in CL patients [50] [51]. Thus, future studies will be performed to determine the ability of these cells to migrate into tissues to determine the contribution of different monocyte subsets to immunopathology at lesion site.

The production of MMP-9 is not spontaneous, being dependent on the interaction of cell to cell, cell to matrix and/or in response to cytokines [52]. As an important inflammatory cytokine TNF is involved in immune regulation and resistance to various microorganisms and exerts a variety of biological activities such as apoptosis, cytotoxicity and induction of MMPs [10,11,12]. We showed here that in CL TNF plays an important role in the regulation of MMP-9 since addition of exogenous TNF enhanced MMP-9 production in HS, whereas neutralization of this cytokine down-regulated MMP-9 synthesis in CL patients. Therefore we propose TNF as a major regulator of MMP-9 production during *L. braziliensis* infection. Our observations agree with what has been previously reported for other diseases [45,46]. For example, in malaria, activation of MMP-9 was dependent upon the

## References

- Llanos Cuentas EA, Cuba CC, Barreto AC, Marsden PD (1984) Clinical characteristics of human Leishmania braziliensis braziliensis infections. Trans -R Soc Trop Med Hyg 78: 845–846.
- Da-Cruz AM, Bertho AL, Oliveira-Neto MP, Coutinho SG (2005) Flow cytometric analysis of cellular infiltrate from American tegumentary leishmaniasis lesions. Br J Dermatol 153: 537–543.
- Kaye P, Scott P (2011) Leishmaniasis: complexity at the host-pathogen interface. Nat Rev Microbiol 9: 604–615.
- Ribeiro-de-Jesus A, Almeida RP, Lessa H, Bacellar O, Carvalho EM (1998) Cytokine profile and pathology in human leishmaniasis. Braz J Med Biol Res 31: 143–148.
- Ziegler-Heitbrock L, Ancuta P, Crowe S, Dalod M, Grau V, et al. (2010) Nomenclature of monocytes and dendritic cells in blood. Blood 116: e74–80.
- Zawada AM, Rogacev KS, Rotter B, Winter P, Marell RR, et al. (2011) SuperSAGE evidence for CD14++CD16+ monocytes as a third monocyte subset. Blood 118: e50–61.
- Ziegler-Heitbrock L (2007) The CD14+ CD16+ blood monocytes: their role in infection and inflammation. J Leukoc Biol 81: 584–592.
- Belge KU, Dayyani F, Horelt A, Siedlar M, Frankenberger M, et al. (2002) The proinflammatory CD14+CD16+DR++ monocytes are a major source of TNF. J Immunol 168: 3536–3542.
- Fingerle G, Pforte A, Passlick B, Blumenstein M, Strobel M, et al. (1993) The novel subset of CD14+/CD16+ blood monocytes is expanded in sepsis patients. Blood 82: 3170–3176.
- Baugh JA, Bucala R (2001) Mechanisms for modulating TNF alpha in immune and inflammatory disease. Curr Opin Drug Discov Devel 4: 635–650.
- Gupta S (2002) A decision between life and death during TNF-alpha-induced signaling. J Clin Immunol 22: 185–194.
- Lehmann W, Edgar CM, Wang K, Cho TJ, Barnes GL, et al. (2005) Tumor necrosis factor alpha (TNF-alpha) coordinately regulates the expression of specific matrix metalloproteinases (MMPS) and angiogenic factors during fracture healing. Bone 36: 300–310.
- Nagase H, Ogata Y, Suzuki K, Enghild JJ, Salvesen G (1991) Substrate specificities and activation mechanisms of matrix metalloproteinases. Biochem Soc Trans 19: 715–718.
- Murphy G, Willenbrock F, Crabbe T, O'Shea M, Ward R, et al. (1994) Regulation of matrix metalloproteinase activity. Ann N Y Acad Sci 732: 31–41.
- Elkington PT, O'Kane CM, Friedland JS (2005) The paradox of matrix metalloproteinases in infectious disease. Clin Exp Immunol 142: 12–20.
- Brunner S, Kim JO, Methe H (2010) Relation of matrix metalloproteinase-9/ tissue inhibitor of metalloproteinase-1 ratio in peripheral circulating CD14+ monocytes to progression of coronary artery disease. Am J Cardiol 105: 429– 434.
- Ladwig GP, Robson MC, Liu R, Kuhn MA, Muir DF, et al. (2002) Ratios of activated matrix metalloproteinase-9 to tissue inhibitor of matrix metalloproteinase-1 in wound fluids are inversely correlated with healing of pressure ulcers. Wound Repair Regen 10: 26–37.

production of TNF and its activation was decrease by anti-TNF treatment [46].

This study brings evidence for the role of monocytes in the pathology associated with *L. braziliensis* infection and suggests that monocytes and MMP-9 may play key roles in tissue destruction. Specifically, the excessive production of TNF by monocytes observed in CL increases the production of MMP-9 leading to an imbalance between production of this enzyme and its inhibitor, TIMP-1. As a consequence, there is an excessive degradation of the basal membrane, migration of inflammatory cells to the site of infection and ulcer development. Thus, therapeutic modulation of MMP-9 may be a useful approach for improving disease outcome in *L. braziliensis* patients.

## Acknowledgments

We thank Cristiano Franco for secretarial assistance.

### **Author Contributions**

Conceived and designed the experiments: TMC STP DM PS EMC LPC. Performed the experiments: TMC STP RSC FON DPB. Analyzed the data: TMC STP FON DPB LPC. Contributed reagents/materials/analysis tools: AQ FON DPB PS EMC. Wrote the paper: TMC STP DPB DM PS EMC LPC.

- Teles RM, Teles RB, Amadeu TP, Moura DF, Mendonca-Lima L, et al. (2010) High matrix metalloproteinase production correlates with immune activation and leukocyte migration in leprosy reactional lesions. Infect Immun 78: 1012– 1021.
- Maretti-Mira AC, de Pinho Rodrigues KM, de Oliveira-Neto MP, Pirmez C, Craft N (2011) MMP-9 activity is induced by Leishmania braziliensis infection and correlates with mucosal leishmaniasis. Acta Trop 119: 160–164.
- Sorokin L (2010) The impact of the extracellular matrix on inflammation. Nat Rev Immunol 10: 712–723.
- Machado PR, Ampuero J, Guimaraes LH, Villasboas L, Rocha AT, et al. (2010) Miltefosine in the treatment of cutaneous leishmaniasis caused by Leishmania braziliensis in Brazil: a randomized and controlled trial. PLoS Negl Trop Dis 4: e912.
- Du P, Kibbe WA, Lin SM (2008) lumi: a pipeline for processing Illumina microarray. Bioinformatics 24: 1547–1548.
- Smyth GK (2004) Linear models and empirical bayes methods for assessing differential expression in microarray experiments. Stat Appl Genet Mol Biol 3: Article3.
- Wettenhall JM, Smyth GK (2004) limmaGUI: a graphical user interface for linear modeling of microarray data. Bioinformatics 20: 3705–3706.
- Reich M, Liefeld T, Gould J, Lerner J, Tamayo P, et al. (2006) GenePattern 2.0. Nat Genet 38: 500–501.
- Reed SG, Badaro R, Lloyd RM (1987) Identification of specific and crossreactive antigens of Leishmania donovani chagasi by human infection sera. J Immunol 138: 1596–1601.
- Bottrel RL, Dutra WO, Martins FA, Gontijo B, Carvalho E, et al. (2001) Flow cytometric determination of cellular sources and frequencies of key cytokineproducing lymphocytes directed against recombinant LACK and soluble Leishmania antigen in human cutaneous leishmaniasis. Infect Immun 69: 3232–3239.
- Murphy G, Nagase H (2008) Progress in matrix metalloproteinase research. Mol Aspects Med 29: 290–308.
- Goetzl EJ, Banda MJ, Leppert D (1996) Matrix metalloproteinases in immunity. J Immunol 156: 1–4.
- Welgus HG, Campbell EJ, Cury JD, Eisen AZ, Senior RM, et al. (1990) Neutral metalloproteinases produced by human mononuclear phagocytes. Enzyme profile, regulation, and expression during cellular development. J Clin Invest 86: 1496–1502.
- Makela M, Salo T, Larjava H (1998) MMP-9 from TNF alpha-stimulated keratinocytes binds to cell membranes and type I collagen: a cause for extended matrix degradation in inflammation? Biochem Biophys Res Commun 253: 325– 335.
- Carvalho LP, Passos S, Schriefer A, Carvalho EM (2012) Protective and pathologic immune responses in human tegumentary leishmaniasis. Front Immunol 3: 301.
- Unger A, O'Neal S, Machado PR, Guimaraes LH, Morgan DJ, et al. (2009) Association of treatment of American cutaneous leishmaniasis prior to ulcer

development with high rate of failure in northeastern Brazil. Am J Trop Med Hyg 80: 574–579.

- Barral A, Guerreiro J, Bomfim G, Correia D, Barral-Netto M, et al. (1995) Lymphadenopathy as the first sign of human cutaneous infection by Leishmania braziliensis. Am J Trop Med Hyg 53: 256–259.
- D'Oliveira A, Jr., Machado P, Bacellar O, Cheng LH, Almeida RP, et al. (2002) Evaluation of IFN-gamma and TNF-alpha as immunological markers of clinical outcome in cutaneous leishmaniasis. Rev Soc Bras Med Trop 35: 7–10.
- Bullard KM, Lund L, Mudgett JS, Mellin TN, Hunt TK, et al. (1999) Impaired wound contraction in stromelysin-1-deficient mice. Ann Surg 230: 260–265.
- Muller M, Trocme C, Lardy B, Morel F, Halimi S, et al. (2008) Matrix metalloproteinases and diabetic foot ulcers: the ratio of MMP-1 to TIMP-1 is a predictor of wound healing. Diabet Med 25: 419–426.
- Costa JD, Nogueira de Melo AC, Vermelho AB, Meirelles Mde N, Porrozzi R (2008) In vitro evidence for metallopeptidase participation in hepatocyte damage induced by Leishmania chagasi-infected macrophages. Acta Trop 106: 175–183.
- Maretti-Mira AC, de Oliveira-Neto MP, Da-Cruz AM, de Oliveira MP, Craft N, et al. (2011) Therapeutic failure in American cutaneous leishmaniasis is associated with gelatinase activity and cytokine expression. Clin Exp Immunol 163: 207–214.
- Faria DR, Souza PE, Duraes FV, Carvalho EM, Gollob KJ, et al. (2009) Recruitment of CD8(+) T cells expressing granzyme A is associated with lesion progression in human cutaneous leishmaniasis. Parasite Immunol 31: 432–439.
- 41. Carvalho LP, Passos S, Bacellar O, Lessa M, Almeida RP, et al. (2007) Differential immune regulation of activated T cells between cutaneous and mucosal leishmaniasis as a model for pathogenesis. Parasite Immunol 29: 251– 258.
- Novais FO, Carvalho LP, Graff JW, Beiting DP, Ruthel G, et al. (2013) Cytotoxic T cells mediate pathology and metastasis in cutaneous leishmaniasis. PLoS Pathog 9: e1003504.

- Rittig MG, Bogdan C (2000) Leishmania-host-cell interaction: complexities and alternative views. Parasitol Today 16: 292–297.
- 44. Giudice A, Vendrame C, Bezerra C, Carvalho LP, Delavechia T, et al. (2012) Macrophages participate in host protection and the disease pathology associated with Leishmania braziliensis infection. BMC Infect Dis 12: 75.
- Price NM, Gilman RH, Uddin J, Recavarren S, Friedland JS (2003) Unopposed matrix metalloproteinase-9 expression in human tuberculous granuloma and the role of TNF-alpha-dependent monocyte networks. J Immunol 171: 5579–5586.
- 46. Prato M, Giribaldi G, Polimeni M, Gallo V, Arese P (2005) Phagocytosis of hemozoin enhances matrix metalloproteinase-9 activity and TNF-alpha production in human monocytes: role of matrix metalloproteinases in the pathogenesis of falciparum malaria. J Immunol 175: 6436–6442.
- Clark RT, Nance JP, Noor S, Wilson EH (2011) T-cell production of matrix metalloproteinases and inhibition of parasite clearance by TIMP-1 during chronic Toxoplasma infection in the brain. ASN Neuro 3: e00049.
- Marzano AV, Cugno M, Trevisan V, Fanoni D, Venegoni L, et al. (2010) Role of inflammatory cells, cytokines and matrix metalloproteinases in neutrophilmediated skin diseases. Clin Exp Immunol 162: 100–107.
- Bittencourt AL, Barral A (1991) Evaluation of the histopathological classifications of American cutaneous and mucocutaneous leishmaniasis. Mem Inst Oswaldo Cruz 86: 51–56.
- Ancuta P, Wang J, Gabuzda D (2006) CD16+ monocytes produce IL-6, CCL2, and matrix metalloproteinase-9 upon interaction with CX3CL1-expressing endothelial cells. J Leukoc Biol 80: 1156–1164.
- Soares G, Barral A, Costa JM, Barral-Netto M, Van Weyenbergh J (2006) CD16+ monocytes in human cutaneous leishmaniasis: increased ex vivo levels and correlation with clinical data. J Leukoc Biol 79: 36–39.
- Nagase H (1997) Activation mechanisms of matrix metalloproteinases. Biol Chem 378: 151–160.